Author's response to reviews

Title: Giant Mediastinal Mature Teratoma With Increased Exocrine Pancreatic Activity

Authors:

Franco Stella M.D. (franco.stella@aosp.bo.it)
Fabio Davoli M.D. (fab_78_16l@hotmail.com)
Alessandro Bini Prof. (abini@aosp.bo.it)

Version: 2 Date: 29 November 2010

Author's response to reviews: see over
AUTHORS’ CONTRIBUTIONS SECTION

FS was the first surgery during the operation, he choose surgical strategy (median sternotomy).

FD was a major contributor in writing the manuscript and collecting all the data about the patient suggesting the particularity of the high levels of Amilase in the cystic fluid.

AB was involved in the pre-operative evaluation and diagnosis of this neoformation and he made the review of the literature.

All authors read and approved the final manuscript.

Dr. Franco Stella
Dr. Fabio Davoli
Prof. Alessandro Bini